Author's response to reviews

Title: Access and utilisation of maternity care for disabled women who experience domestic abuse: a systematic review

Authors:

Jenna P Breckenridge (j.breckenridge@dundee.ac.uk)
John Devaney (j.devaney@qub.ac.uk)
Thilo Kroll (t.kroll@dundee.ac.uk)
Anne Lazenbatt (a.lazenbatt@qub.ac.uk)
Julie Taylor (julie.taylor@ed.ac.uk)
Caroline Bradbury-Jones (caroline.bradbury-jones@manchester.ac.uk)

Version: 3  Date: 30 June 2014

Author's response to reviews: see over
Dear Professor Begley,

Re: MS: 593300441294616 - Access and utilisation of maternity care for disabled women who experience domestic abuse: a systematic review

Thank you for considering the above article for publication in BMC Pregnancy and Childbirth. We have addressed each of the editorial comments and outlined our response point by point in the table below. We hope the paper is satisfactory and now meets the editorial requirements.

Yours sincerely,

Dr Jenna Breckenridge

On behalf of the authors.

| Please include your funding information in the acknowledgements section rather than as a separate section | This has been done. |
| Additional files should be listed separately at the end of the manuscript. | This has been done and the additional files are listed on page 41. The text has also been altered on pages 6 and 9 to direct the reader to these additional files. |
| Figure files should also be listed separately, following the tables. Figure files (such as your flow chart) will appear in the manuscript. | Figure 1 (flow chart) has been included in the main manuscript following the tables on page 40. |
| All files should be listed in numerical order i.e. your tables should be labelled 1-3 and mentioned as such in the manuscript. | This has been done. |